Author's response to reviews

Title: The Role of PET/CT for the Detection of Gastric Cancer Recurrence

Authors:

Sung Hoon Sim (boron@hanmail.net)
Yu Jung Kim (cong1005@medimail.co.kr)
Do-Youn Oh (ohdoyoun@snu.ac.kr)
Se-Hoon Lee (shlee119@snu.ac.kr)
Dong-Wan Kim (dwkimmd@chol.com)
Won Jun Kang (mdkwj@snu.ac.kr)
Seock-Ah Im (moisa@snu.ac.kr)
Tae-You Kim (kimty@snu.ac.kr)
Woo Ho Kim (woohokim@snu.ac.kr)
Dae Seog Heo (heo1013@plaza.snu.ac.kr)
Yung-Jue Bang (bangyj@plaza.snu.ac.kr)

Version: 5 Date: 23 February 2009

Author's response to reviews: see over
February 24, 2009
Editor-in-Chief
BMC cancer

Dear, Editor - in-Chief

We thank the editors and reviewers of the *BMC cancer* for taking their time to review our submitted article entitled “*The Role of PET/CT in Detection of Gastric Cancer Recurrence*”. We made formatting changes and hope this manuscript will be published soon.

Sincerely yours,

Do-Youn Oh, M.D.,Ph D.
Assistant Professor
Department of Internal Medicine
Seoul National University Hospital
Seoul National University College of Medicine
28 Yongondong Chongno-gu
Seoul,110-744, Korea
Tel:82-2-2072-0701
Fax: 82-2-762-9662
E-mail: ohdoyoun@snu.ac.kr